Author's response to reviews

Title: Breast cancer epidemiology according to recognized breast cancer risk factors in the Prostate, Lung, Colorectal and Ovarian (PLCO) Cancer Screening Trial Cohort.

Authors:

James V Lacey Jr (jimlacey@mail.nih.gov)
Aimee R Kreimer (kreimera@mail.nih.gov)
Saundra S Buys (saundra.buys@hci.utah.edu)
Pamela M Marcus (marcusp@mail.nih.gov)
Shih-Chen Chang (Shih-Chen.Chang@astrazeneca.com)
Michael F Leitzmann (michael.leitzmann@klinik.uni-regensburg.de)
Robert N Hoover (hooverrr@mail.nih.gov)
Philip C Prorok (prorokp@mail.nih.gov)
Christine D Berg (bergc@mail.nih.gov)
Patricia Hartge (hartgep@mail.nih.gov)
PLCO Project Team Prostate, Lung, Colorectal and Ovarian Cancer Screening Trial (kreimera@mail.nih.gov)

Version: 8 Date: 17 December 2008

Author’s response to reviews:

I made modifications to the authors affiliations--I had not accounted for previous versus current affiliations. Sorry for the trouble, and thank you, Aimee